

The Influence of Post-accretion Sedimentation on Marine Magnetic Anomalies

R. Granot, J. Dyment

▶ To cite this version:

R. Granot, J. Dyment. The Influence of Post-accretion Sedimentation on Marine Magnetic Anomalies. Geophysical Research Letters, 2019, 46 (9), pp.4645-4652. 10.1029/2019GL082265 . hal-02390143

HAL Id: hal-02390143 https://hal.science/hal-02390143

Submitted on 20 Mar 2021

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Geophysical Research Letters

RESEARCH LETTER

10.1029/2019GL082265

Key Points:

- Remanent magnetization of the extrusive oceanic basalt is partly or completely removed due to abundant post-accretion sedimentation
- This effect should be considered when interpreting marine magnetic anomalies across passive margins and other heavily sedimented areas

Correspondence to:

R. Granot, rgranot@bgu.ac.il

Citation:

Granot, R., & Dyment, J. (2019). The influence of post-accretion sedimentation on marine magnetic anomalies. *Geophysical Research Letters*, 46, 4645–4652. https://doi.org/ 10.1029/2019GL082265

Received 29 JAN 2019 Accepted 17 APR 2019 Accepted article online 23 APR 2019 Published online 13 MAY 2019

The Influence of Post-accretion Sedimentation on Marine Magnetic Anomalies

R. Granot¹ D and J. Dyment²

¹Department of Geological and Environmental Sciences, Ben-Gurion University of the Negev, Beer-Sheva, Israel, ²Institut de Physique du Globe de Paris, CNRS UMR 7154, Sorbonne Paris Cité, Université Paris Diderot, Paris, France

Abstract Marine magnetic anomalies are a critical observation used to investigate seafloor spreading and the transition between oceanic and continental crust at passive margins. However, pronounced post-accretion sedimentation disturbs the thermal state of the crust and therefore alters its remanent magnetization. To study the link between sedimentation and magnetization of the oceanic crust, we built a series of thermomagnetic forward models coupled with different sedimentation histories. We test our approach against observations from the early Cretaceous southern South Atlantic Ocean. Our simulations suggest that, depending on the thickness of post-accretion sediments, the remanent magnetization of the extrusive basalts is partly or completely removed. Therefore, the typical long-wavelength sea surface marine magnetic anomalies observed above oceanic crust covered by a thick sedimentary pile is almost entirely generated by the magnetization of the deeper crustal layers.

Plain Language Summary Marine magnetic anomalies are a critical observation used to investigate seafloor spreading and the transition between oceanic and continental crust at passive margins. Here we show that the accumulation of sediments over the oceanic crust has a thermal effect, which can lead, under certain conditions, to the complete removal of magnetization in the upper oceanic crust. This, in turn, may have important consequences for how we understand passive margins and the accuracy of plate kinematic models.

1. Introduction

The acquisition of thermoremanent magnetization by the igneous oceanic crust as it drifts from the midocean ridges and cools has proven remarkably useful for the investigation of various geodynamic phenomena. The magnetization is predominantly carried by titanomagnetite (Gee & Kent, 2007; Lowrie et al., 1973; Shau et al., 2000) having varying titanium content (Rubin & Sinton, 2007) and hence the blocking temperature by which the magnetization is locked spans over a large range of temperatures (i.e., ~100 up to ~580 °C). Although the magnetic carriers are sensitive to the post-accretion crustal thermal history, the existence of similar lineated marine magnetic anomalies found on conjugate sides of the different oceanic basins (e.g., Cande et al., 1989; Seton et al., 2014) suggests that, to a first degree, crustal remanent magnetization remains virtually intact. For that reason, post-accretion deposition of sedimentary sequences that leads to elevated temperatures within the oceanic crust was thus far mostly ignored in the context of sea surface marine magnetic anomalies. In certain areas, especially where thick sedimentary sequences exist (e.g., at and near passive margins), the thermal effects of the sediments on crustal magnetization could potentially be significant and might help to resolve some of the ambiguities related to the interpretation of continent-ocean boundaries from marine magnetic anomalies (e.g., Eagles et al., 2015).

An ideal test bed for illustrating the possible effect of post-accretion sedimentation on marine magnetic anomalies would be an area that preserves both conjugate flanks of a mid-ocean ridge, which are covered by a markedly different thickness of sedimentary sequences. Any difference in the shape of the conjugate magnetic anomalies, beyond the expected variance due to the different depth to the basement and other local magnetic and spreading conditions (i.e., direction of the ambient geomagnetic field and remanent magnetization, azimuth of lineation, and seafloor spreading rates), which are relatively well known, could reflect the impact of the sedimentary sequences on magnetization. Such an area exists in the early Cretaceous southern South Atlantic, between anomalies M0 and M6 (121 and 127 Ma, respectively; Malinverno et al., 2012) where the conjugate South American and African flanks are covered by 6 to 9 and 2 to 4 km of sediments, respectively (Figure 1; Hoggard et al., 2017). Interestingly, the observed magnetic profiles crossing these two flanks

©2019. American Geophysical Union. All Rights Reserved.





Figure 1. Mesozoic marine magnetic anomalies collected across the flanks of the southern South Atlantic basin. Background gray map is the satellite-derived gravity grid (Sandwell et al., 2014) reconstructed for the 108-Ma time (South America is fixed, rotation angle is taken from Granot & Dyment, 2015). Colored stripes show the sedimentary thicknesses calculated along seismic profiles (Hoggard et al., 2017). Magnetic anomaly profiles (National Center for Environmental Information) are plotted with positive projected northward. Magnetic profiles shown with thick lines are projected in Figure 3. Green lines mark the locations of the M0 and M4 isochrons.

have markedly different shapes that cannot be perfectly matched by the standard rectangular forward magnetic anomaly models (Figures 2 and 3). The short-wavelength (15 to ~30 km) anomalies are mostly missing from the South American anomalies (5% to 15% of the expected power is observed; Figure 2), whereas the short-wavelength spectra of the African anomalies is only partly reduced (20% to 30% of the expected power is observed; Figure 2). To better understand the possible influence of the sediments on the crustal magnetization, we first constructed a series of forward magnetic models based on simulations of magnetization acquisition through time given the different thermal and sediment deposition history of the two flanks. We then compare our results with the observations from the southern South Atlantic basin and show that sedimentation could have an important demagnetization effect on the upper oceanic crust.

2. Thermomagnetic Modeling

A realistic approach for modeling marine magnetic anomalies that takes into account both the thermal evolution of the oceanic lithosphere and the magnetic properties of its constituents has been previously applied to the study of the shape (skewness) of the anomalies (Arkani-Hamed, 1991; Dyment et al., 1997; Dyment & Arkani-Hamed, 1995). In this approach, first the thermal evolution of the oceanic lithosphere is calculated assuming a half-space model where the lithosphere is conductively cooling from the top. To account for the acquisition of both the thermoremanent magnetization and the secondary viscous remanent magnetization, the theoretical blocking curves of single domain titanomagnetite carriers (Pullaiah et al., 1975) are adopted. The models assign, for each time step, a positive or negative magnetization component based on the



Figure 2. Multitaper spectral analysis (Prieto et al., 2009) of the sea surface magnetic anomalies. Thin red and blue lines show the spectra of the 16 African and South American magnetic profiles shown in Figure 1, respectively. Thick lines delineate their averages. Dashed lines show the spectra of the standard forward magnetic anomalies (black lines in Figure 3). A time-bandwidth product of 4 and 8 tapers were used. Due to the band pass Earth filter (Schouten & McCamy, 1972), only wavelengths longer than 15 km can potentially be preserved.

geomagnetic polarity state as being defined for the Cenozoic (Cande & Kent, 1995) and Mesozoic (Malinverno et al., 2012). We modified this thermoviscous remanent magnetization modeling approach to include the thermal effect of the post-accretion sedimentary sequence deposited on top the oceanic crust. We assumed that no internal heat is being generated within the sediments and assigned 0 °C as the thermal boundary condition to the top of the sedimentary layer (i.e., seafloor). For simplicity, we assumed that the temperatures within the sediments and in the top of the oceanic crust reach steady-state conditions immediately after deposition. Since in reality the temperatures are slowly increasing (over $\sim 10^7$ million of years) before they reach the steady-state conditions, our results provide an upper limit, conservative perspective for the influence of sedimentation on magnetization. The sediments are assumed to have negligible natural remanent magnetization (NRM) and thus were not used in the calculation of the anomalies. We assigned saturation NRM values following the compilation of Gee and Kent (2007) whereby the extrusive basalt, sheeted dike, and gabbro layers have NRM values of 6, 2, and 1.5 A/m, respectively (Figure 4). We constructed a series of forward models for which we computed the magnetization and resultant anomalies for the last 140 million years at a time interval of 0.1 Myr, the equivalent of 2.5 km at the half seafloor spreading rate of 25 km/Myr that is appropriate for the southern South Atlantic (Granot & Dyment, 2015). As illustrated by Figure 2, sedimentation may lead to the attenuation of the high frequency content, and we therefore investigate its influence on crustal magnetization, the source of such anomalies, and computed the magnetization down to a 6-km depth below the top of the basalt. The basalt, sheeted dike and gabbro layers were assigned 0.5, 1, and 4.5 km thicknesses, respectively.



Figure 3. Observed and forward-modeled magnetic anomalies shown for the South American (a) and African (b) southern South Atlantic flanks. Bottom profiles show the results from the standard, simple rectangular magnetized prisms forward magnetic modeling. We used 1-km-thick magnetic source layer with magnetization of 8 A/m. Ambient and remanent magnetization directions were based on Thébault et al. (2015) and Torsvik et al. (2012), respectively. The colored profiles are the total magnetic anomalies computed with the coupled thermomagnetic forward modeling approach that takes the effect of post-accretion sedimentation into account. Red and blue magnetic profiles show the expected anomalies when using basalt-blocking temperatures of 200–400 and 100–300 °C, respectively. These profiles were taken from the models shown in Figure 4 (black profiles), where the results were skewed and scaled to take into account the actual ambient and remanent magnetization directions and azimuth of seafloor spreading of the two flanks of the southern South Atlantic. Note that the location of the anomalies shown along these four profiles is displaced outward due to the demagnetization of the upper crust (Figure 4). Therefore, their locations (not their shape), for the purpose of comparison, were shifted backward. Four representative magnetic anomaly profiles (black lines) are shown at the top (for locations see Figure 1).

Accurate knowledge on the characteristics of the sedimentary sequences deposited on top of the early Cretaceous flanks of the southern South Atlantic is required in order to reconstruct its crustal thermal setting hence its magnetization. The global sedimentary thickness grids have significant uncertainties in regions of thick sedimentary sequences, such as those adjacent to continental margins (e.g., Laske & Masters, 1997), therefore we adopted the sedimentary thickness values that were constructed along multichannel seismic reflection profiles and wide-angle experiments (Figure 1; Hoggard et al., 2017). In the regions of abundant sedimentation, such as the studied area, the reported uncertainties on these estimations can reach 400 m. The deposition histories of the South American (Morales et al., 2017) and African (Dressel et al., 2015) flanks, based on seismic and well data, were inserted in the models (Figures 4a and 4b).

The blocking temperature ranges used while modeling the magnetization of the different layers have critical effect on the resultant magnetization. Unlike the deeper gabbros and sheeted dikes where the magnetization is generally carried by magnetite with low titanium content, for which a characteristic blocking temperature range of 400 to 580 °C is the most appropriate, the magnetic carrier of the basaltic lavas is titanomagnetite with variable titanium content (Gee & Kent, 2007). Hence, irregular ranges of blocking temperatures have been observed for different drilled lavas. We therefore constructed a series of models that aim to cover the entire span of the blocking temperatures commonly observed in the upper basaltic samples (Figure 4).





Figure 4. Results from the thermomagnetic forward modeling. (a, b) Sedimentation rate curves of the investigated Cretaceous-aged South American (a; Morales et al., 2017) and African (b; Dressel et al., 2015) southern South Atlantic flanks that were incorporated into the models. (c–f) Magnetization grids and the resultant magnetic anomalies. The anomalies are computed separately for the gabbro, sheeted dike and basalt layers and are shown in light red, purple, and red, respectively. The sum of these three contributions is shown in black. In all models, the blocking temperatures for the sheeted dikes and gabbros range between 400 and 580 °C. The blocking temperatures of the basalts were simulated with ranges of 100 to 300 °C (c, d) and 200 to 400 °C (e, f). These models were constructed using constant 7- and 10-km depths for the African and South American top of the basalt layer, respectively.

The results from the thermomagnetic modeling provide important new insights to the origin of magnetic anomalies on crust with abundant sedimentation. The lower crustal gabbros and the sheeted dikes display similar remanent magnetization on both flanks (Figure 4), regardless of the thickness of the sedimentary sequence deposited on top the oceanic crust. This similarity is due to their high blocking temperature range, which prevents any significant thermal demagnetization. Nevertheless, the resultant magnetic anomalies predicted for these two layers show significant differences with lower amplitudes and higher attenuation of the short-wavelength signal predicted for the South American flank compared to the African flank (Figure 4). These differences reflect the contrasting depths of the sheeted dikes and gabbros, which are, due to the different cumulative water depths and sediment thicknesses (Figure 1), generally 3 km deeper on the South American flank.

G

100

Contrasting differences in the magnetization states of the two flanks are predicted for the basalts, where the magnetic carrier (titanomagnetite) is highly sensitive to the higher temperatures induced by sedimentation. When using the low range of blocking temperatures (100 to 300 °C; Figures 4c and 4d) for the extrusive basalt, the early Cretaceous remanent magnetization of the South American flank is entirely erased and therefore produces no sea surface magnetic anomaly. A similar effect is also observed on the African flank, although at a less developed stage; hence, small amplitude (~20 nT) short-wavelength surface anomalies arise from the extrusive basalt magnetization. When using a higher blocking temperature range (200 to 400 °C; Figures 4e and 4f) for the extrusive basalts, the magnetization of the South American basalts is only partially erased, whereas that of the African ones remains almost intact. Combined with the effect of the different depths to these source layers, the resulting surface anomalies of the African flank are significantly stronger and less attenuated than those of the South American flank.

3. Discussion

Our coupled thermomagnetic models indicate that post-accretion sedimentation might strongly attenuate, under certain conditions, the magnetization of the upper part of the oceanic crust. This effect is governed by the thickness of the sedimentary sequence and modulated by the blocking temperatures of the basalts (Figure 4). The magnetic observations from the South Atlantic early Cretaceous crust (Figures 1 and 2) reveal that the short-wavelength (15 to 30 km) anomaly content is almost entirely missing in most of the South American profiles, and in some profiles, mostly in the African profiles, this signal is strongly subdued. The thermomagnetic models show that indeed the short-wavelength signal is reduced (for blocking temperatures of 200–400 °C) or eliminated (for blocking temperatures of 100–300 °C). The variable short-wavelength content observed along the different magnetic profiles of the South Atlantic (Figures 1–3) may reflect along-axis variations in the titanium content within the lavas (Lattard et al., 2006) in relation to varying degree of fractionation, or a more complex three-dimensional thermal structure at the ridge axis and its subsequent evolution (Gac et al., 2003, 2006), both probably related to ridge segmentation.

As a consequence, the interpretation of marine magnetic anomalies found near passive continental margins, where the crust is commonly covered by a thick sedimentary pile (>~3 km), should be made with special caution. There, the remanent magnetization of crust that formed during short-lived periods of stable geomagnetic polarity (i.e., chrons) may have been partially, or fully, erased, leading to subdued (or missing) short-wavelength magnetic anomalies. Passive continental margins covered by abundant sediments, such as the edges of the North Atlantic (Bronner et al., 2011), eastern Mediterranean (Granot, 2016), and Australian-Antarctic Basin (Tikku & Cande, 1999), commonly reveal abnormally long-wavelength lineated anomalies. The lack of short-wavelength anomalies there could be attributed to synrifting sedimentation (Levi & Riddihough, 1986), to ultraslow spreading rates or to the post-accretion sedimentation shown here. In any case, these magnetic observations have contributed to ambiguities in the interpretation of the nature of the crust (i.e., continental, transitional, or oceanic; Eagles et al., 2015). Our results indicate that the lack of short-wavelength anomalies may in fact be attributed to thermal demagnetization and therefore that the nature of the crust underlying some of these anomalies may well be oceanic.

Finally, the diminution or complete removal of thermoremanent magnetization in the extrusive basalt heighten the relative contribution of the deeper magnetic layers, affecting not only the amplitude and the short-wavelength content of the anomalies but also their anomalous skewness (e.g., Cande, 1976; Dyment et al., 1994), therefore compromising the possibility to use these magnetic anomalies to determine paleomagnetic poles (e.g., Koivisto et al., 2011; Schouten & Cande, 1976). Furthermore, the altered magnetic anomalies affect the picking location of the polarity reversals on the sea surface magnetic anomalies, which are further offset outward from the ridge axis compared to their location if the original remanent magnetization of the extrusive basalt had been preserved. This effect adds to the outward displacement observed on magnetic anomalies worldwide (DeMets & Wilson, 2008), which mostly results from the different geometry and relative contribution of the lower and upper crustal magnetized layers and may partly explain the large variation of this parameter among ocean basin (DeMets & Wilson, 2008). For the South Atlantic crust (half spreading rate of 25 km/Myr), our model predicts up to 30 km of outward offset in the location of the Mesozoic magnetic anomaly picks. Inclusion of such isochrons in the calculation of plate kinematic models could result in false elevated spreading rates and mislocation of the rotation poles.

4. Conclusions

New thermomagnetic forward models that account for the post-accretion sedimentation history have been used to explain the subdued marine magnetic anomalies observed in the southern South Atlantic early Cretaceous flanks. The results indicate that, for crust underlying thick sedimentary sequences, the original magnetization of the extrusive basalt is being eroded, either partly or entirely, leading to the disappearance of the short-wavelength content, reduction in the amplitude and modification of the shape of the sea surface magnetic anomalies. This effect should be taken into account when interpreting marine magnetic anomalies across passive margins and other areas covered by thick sedimentary sequences, commonly used to infer the nature of the crust, to locate paleomagnetic poles, and to calculate plate kinematic models.

References

- Arkani-Hamed, J. (1991). Thermoremanent magnetization of the oceanic lithosphere inferred from a thermal evolution model:
- Implications for the source of marine magnetic anomalies. *Tectonophysics*, 192, 81–96. https://doi.org/10.1016/0040-1951(91)90247-PBronner, A., Sauter, D., Manatschal, G., Péron-Pinvidic, G., & Munschy, M. (2011). Magmatic breakup as an explanation for magnetic anomalies at magma-poor rifted margins. *Nature Geoscience*, 4, 549–553. https://doi.org/10.1038/NGEO1201
- Cande, S. C. (1976). A palaeomagnetic pole from Late Cretaceous marine magnetic anomalies in the Pacific. *Geophysical Journal of the Royal Astronomical Society*, 44, 547–566. https://doi.org/10.1111/j.1365-246X.1976.tb00292.x
- Cande, S. C., & Kent, D. V. (1995). Revised Calibration of the Geomagnetic Polarity Timescale for the Late Cretaceous and Cenozoic. Journal of Geophysical Research, 100, 6093–6095. https://doi.org/10.1029/94jb03098
- Cande, S. C., LaBrecque, J. L., Larson, R. L., Pitman, W. C., Golovchenko, X., & Haxby, W. F. (1989). Magnetic Lineations of the World's cean Basins. Tulsa, Oklahoma: American Association of Petroleum Geologists.
- DeMets, C., & Wilson, D. S. (2008). Toward a minimum change model for recent plate motions: Calibrating seafloor spreading rates for outward displacement. *Geophysical Journal International*, 174, 825–841. https://doi.org/10.1111/j.1365-246X.2008.03836.x
- Dressel, I., Scheck-Wenderoth, M., Cacace, M., Lewerenz, B., Gotze, H. J., & Reichert, C. (2015). Reconstruction of the southwestern African continental margin by backward modeling. *Marine and Petroleum Geology*, 67, 544–555. https://doi.org/10.1016/j. marpetgeo.2015.06.006
- Dyment, J., & Arkani-Hamed, J. (1995). Spreading-rate-dependent magnetization of the oceanic lithosphere inferred from the anomalous skewness of marine magnetic anomalies. *Geophysical Journal International*, 121, 789–804. https://doi.org/10.1111/j.1365-246X.1995. tb06439.x
- Dyment, J., Arkani-Hamed, J., & Ghods, A. (1997). Contribution of serpentinized ultramafics to marine magnetic anomalies at slow and intermediate spreading centres: insights from the shape of the anomalies. *Geophysical Journal International*, 129, 691–701. https://doi. org/10.1111/j.1365-246X.1997.tb04504.x
- Dyment, J., Cande, S. C., & Arkani-Hamed, J. (1994). Skewness of marine magnetic anomalies created between 85 and 40 Ma in the Indian Ocean. *Journal of Geophysical Research*, 99, 24,121–24,134. https://doi.org/10.1029/94jb02061
- Eagles, G., Pérez-Díaz, L., & Scarselli, N. (2015). Getting over continent ocean boundaries. *Earth-Science Reviews*, 151, 244–265. https://doi. org/10.1016/j.earscirev.2015.10.009
- Gac, S., Dyment, J., Tisseau, C., & Goslin, J. (2003). Axial magnetic anomalies over slow-spreading ridge segments: insights from numerical 3-D thermal and physical modelling. *Geophysical Journal International*, 154, 618–632. https://doi.org/10.1046/j.1365-246X.2003.01971.x
- Gac, S., Tisseau, C., Dyment, J., & Goslin, J. (2006). Modelling the thermal evolution of slow-spreading ridge segments and their off-axis geophysical signature. *Geophysical Journal International*, 164, 341–358. https://doi.org/10.1111/j.1365-246X.2005.02844.x
- Gee, J. S., & Kent, D. V. (2007). Source of oceanic magnetic anomalies and the geomagnetic polarity timescale. In M. Kono (Ed.), *Treatise on Geophysics*, (Vol. 5, pp. 455–507). Amsterdam: Elsevier.
- Granot, R. (2016). Palaeozoic oceanic crust preserved beneath the eastern Mediterranean. *Nature Geoscience*, 9, 701–705. https://doi.org/ 10.1038/Ngeo2784
- Granot, R., & Dyment, J. (2015). The Cretaceous opening of the South Atlantic Ocean. Earth and Planetary Science Letters, 414, 156–163. https://doi.org/10.1016/j.epsl.2015.01.015
- Hoggard, M. J., Winterbourne, J., Czarnota, K., & White, N. (2017). Oceanic residual depth measurements, the plate cooling model, and global dynamic topography. *Journal of Geophysical Research: Solid Earth*, 122, 2328–2372. https://doi.org/10.1002/2016jb013457
- Koivisto, E., Gordon, R., Dyment, J., & Arkani-Hamed, J. (2011). The spreading-rate dependence of anomalous skewness of Pacific plate magnetic anomaly 32: Revisited. *Lithosphere*, 3, 371–378. https://doi.org/10.1130/L167.1

Laske, G., & Masters, G. (1997). A global digital map of sediment thickness. Eos, Transactions of the American Geophysical Union, 78, F483.

- Lattard, D., Engelmann, R., Kontny, A., & Sauerzapf, U. (2006). Curie temperatures of synthetic titanomagnetites in the Fe-Ti-O system: Effects of composition, crystal chemistry, and thermomagnetic methods. *Journal of Geophysical Research*, 111, B12S28. https://doi.org/ 10.1029/2006jb004591
- Levi, S., & Riddihough, R. (1986). Why are marine magnetic anomalies suppressed over sedimented spreading centers. *Geology*, 14, 651–654. https://doi.org/10.1130/0091-7613(1986)14<651:Wammas>2.0.Co;2
- Lowrie, W., Lovlie, R., & Opdyke, N. D. (1973). The magnetic properties of Deep Sea Drilling Project basalts from the Atlantic Ocean. Earth and Planetary Science Letters, 17, 338–349.
- Malinverno, A., Hildebrandt, J., Tominaga, M., & Channell, J. E. T. (2012). M-sequence geomagnetic polarity time scale (MHTC12) that steadies global spreading rates and incorporates astrochronology constraints. *Journal of Geophysical Research*, *117*, B06104. https://doi. org/10.1029/2012jb009260
- Morales, E., Chang, H. K., Soto, M., Corrêa, F. S., Veroslavsky, G., de Santa Ana, H., et al. (2017). Tectonic and stratigraphic evolution of the Punta del Este and Pelotas basins (offshore Uruguay). *Petroleum Geoscience*, 23(4), 415–426. https://doi.org/10.1144/petgeo2016-059
- Prieto, G. A., Parker, R. L., & Vernon, F. L. (2009). A Fortran 90 library for multitaper spectrum analysis. *Computers & Geosciences*, 35, 1701–1710. https://doi.org/10.1016/j.cageo.2008.06.007
- Pullaiah, G., Irving, E., Buchan, K. L., & Dunlop, D. J. (1975). Magnetization changes caused by burial and uplift. Earth and Planetary Science Letters, 28, 133–143. https://doi.org/10.1016/0012-821x(75)90221-6

Acknowledgments

This study was funded by ISF grant 716/16 and was initiated when R. G. was invited professor at IPGP. We thank Roger Searle and an anonymous reviewer for their helpful and constructive comments. J. D. thanks Jafar Arkani-Hamed for pointing his attention on the importance of thermal demagnetization at passive margins. This is IPGP contribution 4035. Marine magnetic data can be obtained at the NCEI (GEODAS database https:// www.ngdc.noaa.gov/mgg/geodas/ trackline.html). The authors have no financial conflict of interests.

- Rubin, K. H., & Sinton, J. M. (2007). Inferences on mid-ocean ridge thermal and magmatic structure from MORB compositions. *Earth and Planetary Science Letters*, 260, 257–276. https://doi.org/10.1016/j.epsl.2007.05.035
- Sandwell, D. T., Müller, R. D., Smith, W. H. F., Garcia, E., & Francis, R. (2014). New global marine gravity model from CryoSat-2 and Jason-1 reveals buried tectonic structure. *Science*, 346, 65–67. https://doi.org/10.1126/science.1258213

Schouten, H., & Cande, S. C. (1976). Palaeomagnetic poles from marine magnetic anomalies. Geophysical Journal of the Royal Astronomical Society, 44, 567–575. https://doi.org/10.1111/j.1365-246X.1976.tb00293.x

- Schouten, H., & McCamy, K. (1972). Filtering marine magnetic anomalies. Journal of Geophysical Research, 77, 7089–7099. https://doi.org/ 10.1029/JB077i035p07089
- Seton, M., Whittaker, J. M., Wessel, P., Müller, R. D., DeMets, C., Merkouriev, S., et al. (2014). Community infrastructure and repository for marine magnetic identifications. *Geochemistry, Geophysics, Geosystems*, 15, 1629–1641. https://doi.org/10.1002/2013GC005176
- Shau, Y.-H., Torii, M., Horng, C.-S., & Peacor, D. R. (2000). Subsolidus evolution and alteration of titanomagnetite in ocean ridge basalts from Deep Sea Drilling Project/Ocean Drilling Program Hole 504B, Leg 83: Implications for the timing of magnetization. *Journal of Geophysical Research*, 105, 23,635–23,649. https://doi.org/10.1029/2000JB900191
- Thébault, E., Finlay, C. C., Beggan, C. D., Alken, P., Aubert, J., Barrois, O., et al. (2015). International Geomagnetic Reference Field: the 12th generation. *Earth, Planets and Space*, 67(1), 67–79. https://doi.org/10.1186/s40623-015-0228-9

Tikku, A. A., & Cande, S. C. (1999). The oldest magnetic anomalies in the Australian-Antarctic Basin: are they isochrons? Journal of Geophysical Research, 104, 661–677. https://doi.org/10.1029/1998JB900034

Torsvik, T. H., van der Voo, R., Preeden, U., Mac Niocaill, C., Steinberger, B., Doubrovine, P. V., et al. (2012). Phanerozoic polar wander, palaeogeography and dynamics. *Earth-Science Reviews*, 114(3-4), 325–368. https://doi.org/10.1016/j.earscirev.2012.06.007